Author's response to reviews

Title: The effects of a physical activity counseling program after an exacerbation in COPD: a randomized controlled pilot study.

Authors:

Miek Hornikx (miek.hornikx@faber.kuleuven.be)  
Heleen Demeyer (heleen.demeyer@faber.kuleuven.be)  
Carlos Augusto Camillo (carlos.camillo@faber.kuleuven.be)  
Wim Janssens (wim.janssens@uzleuven.be)  
Thierry Troosters (thierry.troosters@med.kuleuven.be)

Version: 18  
Date: 5 October 2015

Author's response to reviews: see over
Dear Editor,

We sincerely want to thank you for accepting our article “The effects of a physical activity counseling program after an exacerbation in patients with Chronic Obstructive Pulmonary Disease: a randomized controlled pilot study.” We added the comment and we completed the consort checklist.

On behalf of all authors,

Thierry Troosters
1. The first comment was to add the trial registration date in the abstract.

   This was added to the item “trial registration” in the abstract as follows: Trial registration: clinicaltrials.gov NCT02223962. Registered 4 September 2013.

2. The consort checklist was asked to be completed.

   This form was completed and was added in the online submission.